Sampling notes, La Perouse, 2017-09, 31 Aug – 12 September 2017
CTD/Rosettes:
115 files, 0002 to 0226.
0 stand-alone CTD.

69 rosette casts, 30 casts with no Niskin closed, 1 cast with 1 Niskin just for test. 
0 cast needed two files.
CTD 550, Deck Unit 0425
All cast done mid-ship station

Seasave version 7.26.7.107
Configuration file:
We started with: 2017-09-start.xmlcon.
Because of problems with one monitor cable on the CTD laptop some casts were done with config file 2017-08-start.xmlcon.  Problem fixed after cast 0044 – LC11.

Standard procedure:

The rosette was brought to the surface.  Pumps were turned ON.  Then the rosette was brought down to 10m and back up.  Data started to be archived then, with the rosette at the surface for 30 seconds longer.  Then the cast would start. 

Problems with CTD:

Cast 0036 – LB13: looks like CTD came out of the water after the “down 10 and back”. 
Cast 0104 – LJ06: still archiving at end when out of the water.

Cast 0110 – LBP8: CTD came out of the water after the “down 10 and back”, pressure spikes. 

Cast 0145 – CS04: pressure spikes on the way up during the 50 dbar 30-sec wait. 

Cast 0153 – CS1B: pressure spikes at the bottom and a few on the upcast. 

Niskins & Samples at CTD/Rosette casts:
Cast 0044 – LC11: Niskin 19 (s/n 132) ran out of water so no salinity sample. 

Cast 0051 – LC08: Niskin 2 closed at 175 instead of 200 because shallower than expected.

Cast 0064 – LC01: Niskin 4 (s/n 193) closed at 40 instead of 30.  
Cast 0078 – LD11: Niskin 2 closed at wrong depth, use Niskin 21 for 1500 dbar sample (#237).  Niskin 2 not needed. 

Cast 0124 – LBP2: Niskin 13 closed just for test, not needed, no file needed. 

Cast 0171 – SS3: Niskin 1 closed at wrong depth (281), Niskin 4 used for 20 dbar sample instead.  Niskin 1 not needed in file. 

Cast 0226 – 41: Niskin 3 closed by mistake at 200 dbar.  All other depths are off by one Niskin (so 0 dbar is Niskin 12 instead of 11).  Niskin 3 not needed in the CHE file.
File names – Headers:

Cast 2 – Haro59: file 0002.hex.  
Cast 8 – LB01: file was named 0002..hex when archiving.  After the cast was completed file 0002.hex (Haro59) was renamed 0008.hex.   So file -0008.hex has to be renamed 0002.hex since it is actually Haro59.  All 4 files 0002..xxx have to be renamed 0008.xxx.  

Casts 9 and 10 also have the double dots. 

File 0130 – LQ03: this should be file 0131 (event 130 was drifter deployment).

Cast 0149 – CS3B: Station called CS05 in the header.  
Casts with no PAR sensor:
All casts had the PAR sensor on. 
Casts out of order: 
0078 – 0171 - 
Loops:

Loop 5009 – LB02: 2xChl, 2xOxy, 1xSal

Loop 5037 – LB14: 2xChl, 2xOxy, 1xSal

Loop 5044 – LC11: 2xChl, 2xOxy, 1xSal

Loop 5055 – LC06: 2xChl, 2xOxy, 1xSal

Loop 5072 – LD06: 2xChl, 2xOxy, 1xSal, 1xNut

Loop 5090 – LG04: 2xChl, 2xOxy, 1xSal, 1xNut
Loop 5121 – LBP3: 2xChl, 2xOxy, 1xSal, 1xNut.  Might have been called Loop 5120.
Loop 5156 – CS06: 2xChl, 2xOxy, 1xSal, 1xNut
Loop 5177 – NL01: 2xChl, 2xOxy, 1xSal, 1xNut

Loop 5188 – SS5: 2xChl, 2xOxy, 1xSal, 1xNut
TSG: 
2 files.
Fluorescence at the of file 0002 went way up, kept increasing (last day). 
